Author's response to reviews

Title: Health education for microcredit clients in Peru: a randomized controlled trial

Authors:

Rita Hamad (hamad@post.harvard.edu)
Lia C.H. Fernald (fernal@berkeley.edu)
Dean S. Karlan (dean.karlan@yale.edu)

Version: 6 Date: 21 January 2011

Author's response to reviews: see over
Please find enclosed a revised version of the manuscript. The recommended changes were addressed as follows:

Major Revisions
1. Title page: The format of the affiliations was revised to comply with your request.
2. Additional files: These were referenced in the “Methods” section, and described again in the “Additional Files” section at the end of the manuscript.
3. Typography: This was reviewed and no other edits were made.